Author's response to reviews

Title: Glioblastoma Multiforme in Klippel-Trenaunay-Weber Syndrome: a case report

Authors:

Tevfik Yilmaz (kartaltevfik@hotmail.com)
Ulas Çikla (ulas.cikla@yahoo.com)
Alice Kirst (alice.kirst@gmail.com)
Mustafa Kemal Baskaya (m.baskaya@neurosurg.wisc.edu)

Version: 2
Date: 18 September 2014

Author's response to reviews: see over
September 19, 2014

Michael Kidd, MD
Editor-in Chief
Journal of Medical Case Reports

Dear Dr. Kidd:

Enclosed please find a manuscript entitled “Glioblastoma Multiforme in Klippel-Trenaunay-Weber Syndrome: a case report” and 4 figure submitted for your consideration for publication in *Journal of Medical Case Reports*.

This manuscript has not been published before, either as a whole or in part, nor is it under consideration for publication elsewhere. To best of our knowledge, there have been no report in the English medical literature regarding a KTWS associated with Glioblastoma Multiforme.

As the authors of this manuscript we do not report any conflict of interest related to the manuscript which was not financially supported by any person or institute.

Please address correspondence concerning the manuscript to Mustafa K. Baskaya, MD, Department of Neurological Surgery, University of Wisconsin-Madison, CSC K4/828, 600 Highland Avenue, Madison, WI 53705, USA. Telephone: 608-262 7303; fax: 608-263 1728; e-mail: m.baskaya@neurosurgery.wisc.edu

Thank you for your time and consideration of this manuscript.

Sincerely,

Mustafa K. Baskaya, MD
Associate Professor
Director of Skull Base Surgery
Department of Neurological Surgery
University of Wisconsin Medical School
CSC K4/828
600 Highland Avenue
Madison, WI 53792